Author’s response to reviews

Title: A study of the relationships between KLF2 polymorphisms and body weight control in a French population

Authors:

Aline Meirhaeghe (aline.meirhaeghe-hurez@pasteur-lille.fr)
Dominique Cottel (dominique.cottel@pasteur-lille.fr)
Philippe Amouyel (philippe.amouyel@pasteur-lille.fr)

Version: 8 Date: 13 March 2006

Author’s response to reviews:

I removed the keywords, the running title and put the website address in the reference list using the Reference Manager software as requested.